## **Research Paper:** The Impact of Plasmodium Berghei Exposure In-utero on Neurobehavioral Profile in Mice



Akhabue Keneth Okojie<sup>1,2,3\*</sup> 📵, Khalid Rauf<sup>2</sup> 🕘, Eghosa Iyare<sup>1</sup> 😳

- 1. Reproductive and Developmental Programming Research Group, Department of Physiology, Enugu Campus, University of Nigeria, Enugu, Nigeria.
- 2. Department of Pharmacy, COMSATS Institute of Information Technology, Abbottabad, Pakistan.

3. Department of Physiology, Edo University Iyamho, Uzairue, Edo State, Nigeria.



Citation Okojie, A. K., Rauf, Kh., & Iyare, E. (2019). The Impact of Plasmodium Berghei Exposure In-utero on Neurobehavioral Profile in Mice. *Basic and Clinical Neuroscience*, 10(2), 99-108. http://dx.doi.org/10.32598/bcn.9.10.95

dei http://dx.doi.org/10.32598/bcn.9.10.95

## 00

Article info: Received: 06 Feb 2018 First Revision: 27 Feb 2018 Accepted: 26 May 2018 Available Online: 01 Mar 2019

#### **Keywords:**

Plasmodium berghei, Malaria, In-Utero, Anxiety, Obsessive-Compulsive Disorder

## **ABSTRACT**

**Introduction:** The World Health Organization estimates that about 25 million pregnant mothers are currently at risk for malaria, and that malaria accounts for over 10,000 maternal and 200,000 neonatal deaths per year. The current hypothesis of early life programming supports the premise that many developmental delay and disorders may have their origin Inutero. Therefore, the current study aimed at evaluating the possible impact of experimental malaria exposure In-utero on neurobehavioral profile in mice offspring.

**Methods:** Pregnant mice were intraperitoneally infected on gestational day 13 with  $1.02 \times 10^5$  infected red blood cells. Pregnant mice (both infected and uninfected) were allowed to deliver and the offspring were monitored up to postnatal day 42 when anxiety-like, Obsessive-Compulsive Disorder (OCD), and locomotor activity were evaluated using elevated plus maze, marble burying, and Open Field Test, respectively.

**Results:** The current study showed that maternal infection with Plasmodium berghei resulted in an interesting behavior in offspring characterized by increased anxiety-like and OCD behaviors. Locomotor activity was however not affected.

**Conclusion:** It may be concluded that In-utero exposure to experimental malaria in mice causes behavioral changes.

\* Corresponding Author:

Akhabue Keneth Okojie, PhD.

*Address:* Department of Physiology, Edo University Iyamho, Iyamho-Uzairue, Edo State, Nigeria. *Tel:* +234 (806) 3762090 *E-mail:* okojie.akhabue@edouniversity.edu.ng

.....

## Highlights

- In-Utero plasmodium berghei infection induces anxiety-like behavior in the offspring.
- In-Utero plasmodium berghei infection induces obsessive-compulsive disorder in the offspring.
- In-Utero plasmodium berghei infection does not have any effect on locomotor activity in the offspring.

## Plain Language Summary

The World Health Organization estimates that about 25 million pregnant mothers are currently at risk of inflicting malaria. Also, malaria accounts for over 10000 maternal and 200000 neonatal deaths per year mainly in regions where malaria infection is very high. First-time pregnant women could develop the clinical signs of malaria whereas females who had given birth before with greater immunity may be asymptomatic and therefore not seek treatment, leading to a greater placental parasite load. It could result in significant problems for the developing fetus, especially the brain of the fetus. Therefore, we used experimental malaria in pregnant animals to see how malaria will affect the behavior of offspring. We found out that baby of pregnant animals that had malaria exposure in the uterus, presented with different behavioral problems like anxiety, obsessive-compulsive disorder when the babies were 42 days old. These behavioral changes probably implicate that asymptomatic pregnant mother may potentially expose her fetus to the chronic disease.

## **1. Introduction**

he World Health Organization reported that about roughly 25 million pregnant mothers are at risk for malaria, which accounts for more than 10,000 maternal and 200,000 neonatal deaths each year (Hartman, Rogerson, & Fischer, 2010). These figures could

undervalue the role of malaria on maternal morbidity and mortality. A research in Mozambique on the causes of maternal death through autopsy evaluation found that about 10% of maternal deaths were directly attributed to malaria (Menendez et al., 2008). This implies that in regions in which malaria is endemic, it might directly lead to nearly 25% of maternal deaths. Malaria in pregnancy significantly contributes to perinatal morbidity and mortality. Infection is known to induce high levels of miscarriage, intrauterine passing, premature birth, low-birthweight neonates, and neonatal death (Ofori et al., 2009).

Expecting mothers are three times more at risk for acute disease due to malaria infection in comparison with their non-expecting counterparts, and also have a mortality rate of nearly 50% due to acute disease (Monif & Baker, 2004). It is hypothesized that a vast majority of consequences in pregnancy outcomes results from two fundamental determinants: the immune compromised condition of pregnancy and placental sequestration of the infected erythrocytes. During pregnancy, sequestration of malaria infected Red Blood Cells (iRBC) occurs through the placenta and causes acute anemia as a consequence of the disease in the expecting mothers.

Interestingly, extreme level of placental infection is observed in gravid females with the maximum degree of immunity resulting in moderate maternal manifestation as well as an inordinate growth in fetal complications (Monif & Baker, 2004). Therefore, it might be hypothesized that though primigravidas could develop the clinical signs of malaria, females with greater immunity may be asymptomatic and therefore not seek treatment, leading to a greater placental parasite load. Fetal complications include the placental inflammation, maternal anemia, stillbirth, intrauterine growth restriction, and Low Birth Weight (LBW) neonates.

LBW, as a consequence of preterm delivery or fetal growth restriction, is associated with high risk of developmental setback in children (Allen, 2008; Procianoy, Koch, & Silveira, 2009). In addition, malnutrition and poverty are also considered as risk factors for developmental delay. However, many studies demonstrate that early exposure to diseases such as malaria might have significant deleterious effects on nervous system development. In-utero exposure to infections is associated with increased risk of neurodevelopmental disorders in offspring (O'Callaghan et al., 1994); Nevertheless, there is lack of information regarding the impact of malaria infection during pregnancy on neurobehavioral and cognitive functions of the babies exposed prenatally. Recent studies thus suggest variables that interfere with the closely controlled In-utero environment may alter the normal nervous system developmental processes (Bilbo & Schwarz, 2009; Bale et al., 2010). Current hypothesis of early life programming buttresses the assumption that lots of developmental delay and ailments might have their source In-utero (Okojie et al., 2017; Iyare & Iyare 2006; Iyare & Adegoke 2008; Iyare & Iyare 2007; Bale et al., 2010).

Epidemiological studies also associate maternal infections with greater threat of developmental setback, schizophrenia, autism, and periventricular leukomalacia; a major cause of cognitive impairment in LBW babies (Ellman & Susser 2009). Murine models also show that offspring exposed to maternal infections In-utero experience long-term behavioral deficits including modified gene expression in the brain, altered neurotransmitter physiology, and regional cerebral atrophy (Shi, Fatemi, Sidwell, & Patterson, 2003; Nyffeler, Meyer, Yee, Feldon, & Knuesel, et al., 2006; Ozawa et al., 2006; Gilmore, Jarskog, Vadlamudi, & Lauder, 2004; Golan, Lev, Hallak, Sorokin, & Huleihel, 2005; Meyer et al., 2006), which were however not associated with congenital infection of the fetus. According to increased funding to fight both malaria and maternal mortality, the knowledge about In-utero malaria infection precisely alters offspring development and is vital in the efforts to improve maternal and perinatal health and suppress the spread of the preventable infectious diseases. Hence, the current study aimed at assessing the potential effect of experimental malaria exposure In-utero on neurobehavioral profile in mice offspring.

### 2. Methods

#### 2.1. Animals and parasites

The BALB/c mice were obtained from National Institute of Health, Pakistan, and maintained in conventional housing at COMSATS Institute of Information Technology. Infection experiments were conducted in adult females with 10-12 weeks of age. Plasmodium berghei was a gift from Dr. Aina (Nigeria Institute for Medical Research, Lagos, Nigeria). The iRBC used in the experimental infections were obtained from in vivo passage in BALB/c mice, when the percentage of iRBC was roughly 10%. Thin blood smears were collected daily and stained with Giemsa to monitor parasitemia. All animals were fed with normal rat chow and all procedures conformed to the national guidelines on animal experimentation and welfare and were approved by Comsats Institute of Information Technology.

#### 2.2. Gestation timing and pregnancy monitoring

Presence of vaginal plug and increase of body weight were collectively utilized to determine the time of pregnancy, according to Freyre et al. (2006) two female mice were placed together with a male mouse for two days, and the presence of vaginal plug was controlled each morning. The day of locating the vaginal plug was regarded as Gestation Day one (GD1) and pregnancy development was monitored every day by weighing the females. Successful fertilization was proven from G10 to G13 once the female mouse had a mean weight gain of 3-4 g. Therefore, weight increase was taken as the indication of pregnancy and sudden weight loss as the index of pregnancy loss or interference.

#### 2.3. Pregnancy experimental infection

Pregnant mice were Intraperitoneally (IP) inoculated on GD13 with 1.02×105 iRBC, and parasitemia was recorded every day. The model employed in the current study was predicated on a formerly supported murine model of Malaria in Pregnancy (MIP), which reproduced key pathogenic factors of MIP (Neres, Marinho, Gonçalves, Catarino, & Penha-Gonçalves, 2008). A lower dose of inoculum 105 P. berghei infected erythrocytes, compared with a 10<sup>6</sup> dose utilized in the current study to get rid of a very low birth weight phenotype and raise the amount of live births. Non-infected pregnant mice were used as controls. Pregnant mice (both control and experimental) were permitted to deliver and their offspring were monitored up to the Postnatal Day (PND) 42. Since P. berghei is deadly in BALB/c mice, foster mothers were utilized in the newborn post-natal follow-up studies. Consequently, progenies both from the infected mothers and control ones were additionally transferred to the foster mothers to prevent weight bias due to differential maternal nourishment. The progenies were weighed each day.

#### 2.4. Neurobehavioural assay

#### 2.4.1. Open Field Test (OFT)

Each mouse was separately positioned in the middle of a white acrylic cage  $(30 \times 30 \times 15 \text{ cm})$  and was allowed to explore it for 30 minutes. In this time period, total distance travelled, number of rearing (standing on hind legs with paws pushed against the walls of the stadium), and episodes of grooming were recorded. At the conclusion of testing, the amount of fecal boli was determined and the cage was cleaned using a 10% ethanol solution. In this protocol, the locomotor activity was signified by total distance travelled in the cage, while the vertical display was assigned by the number of rearing. In regards to defecation in male mouse, under certain conditions, this parameter seemed to signify not an emotional reaction but a kind of scent marking (Archer, 1973; Walsh & Cummins 1976).

#### 2.4.2. Marble Burying Test (MBT)

Both in laboratory and natural environments, mice use accessible bedding material to cover obnoxious sources of distress existing in their home surroundings (Archer, Fredriksson, Lewander, & SÖDerberg, 1987). Within this protocol, 10 glasses of marble were equally dispersed in the plastic cage ( $35 \times 50 \times 35$  cm) in the presence of a mouse. It was estimated that after 30 minutes, about two-thirds of the marbles were concealed.

### 2.4.3. Elevated Plus Maze Test (EPMT)

Elevated plus maze reflects a conflict between the rodents' preference for protected areas and their innate motivation to explore novel environments (Pellow, Chopin, File, & Briley, 1985). An apparatus consisting of four arms (30×5 cm) was placed 50 cm above the ground. Two opposite arms were delimited by acrylic vertical walls; whereas the other two opposite arms had unprotected edges (open arms). Mice were placed in the center of maze and allowed to move freely for five minutes. During this period, the cumulative time and frequency of entries into open and closed arms were recorded. An arm entry was defined as the entry of the four paws into an arm. Then, the percentage of entries in open arms, percentage of the time spent in open arms, and the time spent in central platform were calculated. The anxietylike behavior is linked to these last parameters.

Elevated plus maze reflects a battle between the rodents' desire for secure areas and their inborn motivation to search novel surroundings (Pellow et al., 1985). The apparatus consisting of four arms  $(30 \times 5 \text{ cm})$  was placed 50 cm above the ground. The two opposite arms were delimited by acrylic vertical walls; whereas the other two arms had unprotected borders (open arms). Mice were placed in the middle of maze and permitted to move freely for five minutes. In this time, the cumulative frequency and time of entries into open and closed arms were recorded. An arm entrance was set as the entrance of four paws to an arm. Afterward, the proportion of admissions in open arms, portion of time spent in open arms and also the time spent in the central platform were computed. The anxiety-like behavior is connected to those last parameters.

#### 2.5. Statistical analysis

All results are presented as a Means±Standard Error of Mean (SEM). Intergroup comparisons were evaluated with the Student unpaired-test. Differences were considered statistically significant when P<0.05. GraphPad Prism 5 (La Jolla, CA) was used for all statistical analyses.

## **3. Results**

#### 3.1. Blood parasiteamia

It was observed that exposure to P. beghei during pregnancy from GD13 led to gradual increase in the percentage of iRBCs as shown in Figure 1.

#### 3.2. Open Field Test

In this test, no significant difference was observed in locomotor (P=0.6803) and vertical activities (P=0.6671) between the control and experimental groups. Likewise,



#### % Parasiteamia (% iRBCs)

Figure 1. Maternal parasitaemia (GD14-20, n=7) shown as the percent of iRBCs

NEUR<sup>3</sup>SCIENCE



Figure 2. Evaluation of locomotor activity and anxiety-like behavior in open-field test in the offspring of mouse exposed to P. berghei In-utero

a) Total distance covered; b) Rearing behavior; c) Grooming behavior; d) Number of fecal boli; Each bar represents Mean $\pm$ SEM of the data (n=5); \*P<0.05 compared with the control.

there was also no difference in episodes of grooming (P=0.4007). Nevertheless, the mice of P. berghei-exposed group had higher number of fecal pellets  $(5.0\pm1.1 \text{ vs. } 9.4\pm0.8, \text{ P}=0.0106, \text{ Figure 2}).$ 

#### 3.3. Marble Burying Test (MBT)

It was observed that exposure to P. berghei during pregnancy induced a significant increase in the amount of marbles buried in the experimental group, compared with that of the controls  $(2.3\pm0.5 \text{ vs}. 3.6\pm0.4, P=0.043, Figure 3)$ .

#### 3.4. Elevated Plus Maze Test (EPMT)

With the Elevated Plus Maze Test, the group exposed to P. berghei spent different time in the closed arm (240.8 $\pm$ 19.6 vs. 330.8 $\pm$ 13.9 seconds, P=0.0095, Figure 4) and in the open arm (30.0 $\pm$ 2.5 vs. 21.75 $\pm$ 1.6 seconds, P=0.0157, Figure 4). There was no statistical difference

in other anxiety-related parameters such as entries in closed arm (P=0.9481), entries in open arm (P=0.8394), or time in central platform (P=0.1169).

#### 4. Discussion

Despite the fact that malaria infection during pregnancy is very common in sub-Sahara regions, there are very limited studies that demonstrate the effects of maternal infection with malaria in the offspring's behaviors. McDonald et al. (2015) showed that In-utero exposure to maternal malaria infection could change the cognitive and neurological development of offspring. Results of the current study showed that maternal infection with P. berghei resulted in an interesting behavior of offspring characterized by increased anxiety-like and OCD behaviors.



Figure 3. Assessment of anxiety-/OCD-like behavior in MBT in pups of mouse infected with P. berghei during pregnancy

Each bar represents Mean±SEM of the data (n=8).

The offspring exposed to P. berghei were neither congenitally infected nor presented with any phenotypic adverse outcomes associated with MIP. Malaria during pregnancy is marked with the aggregation of iRBCs and monocytes/macrophages in the intervillous space, leading to a specific local immune response in the placenta (Umbers, Aitken, & Rogerson, 2011). The current study results showed a persistent, but gradual increase in the Parasitized Erythrocytes (PEs). Nonetheless, the 10<sup>5</sup> inoculum was correlated with reduced maternal peripheral parasiteamia compared with that of the Neres et al. (2008) using a dose of 10<sup>6</sup> PEs. These results suggested that similar to humans, pregnant mice show increased susceptibility to malaria infection that may affect their progeny or compromise pregnancy, since the current study observed a persistent, but gradual increase in PEs.

The current study observed that P. berghei infection during pregnancy encouraged anxiogenic response in Open Field Test (increased defecation and rearing), and in Elevated Plus Maze Test (reduced percentage of time in open arms and increased time in closed arm). It is logical that the above observations could be a result of distortion of the neuromodulatory systems within the cerebral cortex known to influence behavior. An obvious anxiogenic effect of In-utero malaria infection is that it provides the opportunity to assume that it is associated with a rapid decrease in the amount of serotonin (5-HT) in different cerebral structures. It was previously observed that the case of the cerebral serotonergic system firmly corresponded with the level of stress in rodents and humans (Montegomery, 1991; Griebel, Moreau, Jenck, Misslin, & Martin, 1994).

Grooming activity non-significantly reduced in the experimental group compared with that of the control. Previous studies showed that selective agonists of dopamine D1 and D5 receptors evoked the escalation of grooming behavior, while D2 receptor agonists considerably curtailed this type of behavioral activity (Briley, Chopin, & Moret, 1990; Ferrari, Pelloni, & Giuliani, 1992). Thus, a selective action to some extent as a result of In-utero malaria infection on particular kinds of dopamine receptors should be presumed.

The current study results also showed that experimental malaria infection in mice offspring resulted in increased marble burying, which is a test to evaluate OCD in rodents. It might also reflect the elements of defensive behavior; rodents try to stop predators such as snakes from penetrating their burrows by digging soil and kicking it toward the intruder. Defensive burying is a renowned paradigm to measure anxiety (Pinel & Treit, 1978). The burrowing paradigm proved a sensitive assay to track the growth of prion disease in mice (Deacon, Raley, Perry, & Rawlins, 2001; Guenther, Deacon, Perry, & Rawlins, 2001). The neuronal circuitry of the specific behavior is not yet definitely understood. Thomas et al. (2009) claimed that marble burying reflects a persistent and perseverative behavior over novelty-induced anxiety. The hippocampus and septum are likely to be significant, since hippocampal cytotoxic lesions largely reduce burrowing (Deacon, Croucher, & Rawlins, 2002), and medial prefrontal cortex lesions cause a more compact reduction (Deacon, Penny, & Rawlins, 2003). Some studies also demonstrated that lots of agents including psychostimulants inhibit marble burying (Millan, Girardon, Mullot, Brocco, & Dekeyne, 2002; Wicke & Gross 2005).

Therefore, the current study results demonstrated that experimental In-utero malaria infection possessed a significant modulatory behavioral potential. Experimental In-utero malaria infection resulted in anxiogenic, escape, and OCD behaviors in mice offspring. These behavioral shifts probably show that the implications of asymptomatic pregnant mother may potentially expose her fetus to the chronic disease. It is therefore imperative that all pregnant mothers in malaria endemic regions be screened for malaria parasite and given safe anti-malaria drugs such as prophylaxis. At the same time, the probable mechanisms underlying how In-utero malaria infection affects the cerebral modulatory systems (this kind of action based on the available information, appears to be quite likely) may need specific thorough investigations.

# Basic and Clinical



#### NEUR<sup>3</sup>SCIENCE

Figure 4. Evaluation of neurobehavioral profile in the offspring of mouse exposed to P. berghei infection In-utero using Elevated Plus Maze Test

a) Time spent in open arms; b) Time spent in closed arms; c) Entries in open arms; d) Entries in closed arms; and e) Time spent in central platform; Each bar represents Mean±SEM (n=5); \*\*P<0.0095; \*P<0.0157 compared with controls.

## **Ethical Considerations**

#### Compliance with ethical guidelines

The study protocol was approved by Ethics Committee of Department of Pharmacy, CIIT, Abbottabad Campus (Ethical code: PHM.ETH/FA.16-1116-CIIT-ATD).

#### Funding

The Comsats Institute of Information Technology and the World Academy of Science (CIIT-TWAS) has funded the study through granting CIIT-TWAS sandwich fellowship to the first author.

#### **Authors contributions**

Conceptualization, formal analysis, and writing the draft: Akhabue Kenneth Okojie and Eghosa Iyare; Investigation and writing the draft: Akhabue Kenneth Okojie; Data curation and methodology: Akhabue Kenneth Okojie and Khalid Rauf; Writing-review, editing, visualization, supervision, project administration: Eghosa Iyare and Khalid Rauf; and Funding acquisition: Akhabue Kenneth Okojie.

#### Conflict of interest

The authors declared no conflict of interest.

#### Acknowledgements

Authors would like to offer their special appreciation to Dr. O.O. Aina, Malaria Research Group, Nigerian Institute of Medical Research, for gifting the Plasmodium berghei used for the study.

#### References

- Allen, M. C. (2008). Neurodevelopmental outcomes of preterm infants. *Current Opinions in Neurology*, 21(2), 123-8. [DOI:10.1097/WCO.0b013e3282f88bb4] [PMID]
- Archer, J. (1973). Tests for emotionality in rats and mice: A review. *Animal Behaviour*, 21(2), 205-35. [DOI:10.1016/S0003-3472(73)80065-X]
- Archer, T., Fredriksson, A., Lewander, T., & SÖDerberg, U. L. F. (1987). Marble burying and spontaneous motor activity in mice: Interactions over days and the effect of diazepam. *Scandinavian Journal of Psychology*, 28(3), 242-9. [DOI:10.1111/j.1467-9450.1987.tb00761.x]
- Bale, T. L., Baram, T. Z., Brown, A. S., Goldstein, J. M., Insel, T. R., McCarthy, M. M., et al. (2010). Early life programming and

neurodevelopmental disorders. *Biological Psychiatry*, 68(4), 314-9. [DOI:10.1016/j.biopsych.2010.05.028] [PMID] [PMCID]

- Bilbo, S. D., & Schwarz, J. M. (2009). Early-life programming of later-life brain and behavior: A critical role for the immune system. *Frontiers in Behavioral Neuroscience*, 3(14), 1-14. [DOI:10.3389/neuro.08.014.2009] [PMID] [PMCID]
- Briley, M., Chopin, P., & Moret, C. (1990). Effect of serotonergic lesion on "anxious" behaviour measured in the elevated plus-maze test in the rat. *Psychopharmacology*, 101(2), 187-9. [DOI:10.1007/BF02244124]
- Deacon, R. M., Penny, C., & Rawlins, J. N. P. (2003). Effects of medial prefrontal cortex cytotoxic lesions in mice. *Behavioural Brain Research*, 139(1-2), 139-55. [DOI:10.1016/S0166-4328(02)00225-5]
- Deacon, R. M., Croucher, A., & Rawlins, J. N. P. (2002). Hippocampal cytotoxic lesion effects on species-typical behaviours in mice. *Behavioural Brain Research*, 132(2), 203-13. [DOI:10.1016/S0166-4328(01)00401-6]
- Deacon, R. M., Raley, J. M., Perry, V. H., & Rawlins, J. N. P. (2001). Burrowing into prion disease. *Neuroreport*, *12*(9), 2053-7.[DOI:10.1097/00001756-200107030-00052]
- Ellman, L. M., & Susser, E. S. (2009). The promise of epidemiologic studies: Neuroimmune mechanisms in the etiologies of brain disorders. *Neuron*, 64(1), 25-7. [DOI:10.1016/j.neuron.2009.09.024] [PMID]
- Ferrari, F., Pelloni, F., & Giuliani, D. (1992). Suppressive effect of the dopamine D2 receptor agonist B-HT 920 on rat grooming. *European Journal of Pharmacology*, 216(3), 345-50. [DOI:10.1016/0014-2999(92)90429-8]
- Freyre, A., Falcon, J., Mendez, J., Rodriguez, A., Correa, L., & González, M. (2006). Refinement of the mouse model of congenital toxoplasmosis. *Experimental Parasitology*, 113(3), 154-60. [DOI:10.1016/j.exppara.2005.12.019]
- Gilmore, J. H., Jarskog, L. F., Vadlamudi, S., & Lauder, J. M. (2004). Prenatal infection and risk for schizophrenia: IL-1β, IL-6, and TNFα inhibit cortical neuron dendrite development. *Neuropsychopharmacology*, 29(7), 1221-9. [DOI:10.1038/ sj.npp.1300446] [PMID]
- Golan, H. M., Lev, V., Hallak, M., Sorokin, Y., & Huleihel, M. (2005). Specific neurodevelopmental damage in mice offspring following maternal inflammation during pregnancy. *Neuropharmacology*, 48(6), 903-17. [DOI:10.1016/j.neuropharm.2004.12.023] [PMID]
- Griebel, G., Moreau, J. L., Jenck, F., Misslin, R., & Martin, J. R. (1994). Acute and chronic treatment with 5-HT reuptake inhibitors differentially modulate emotional responses in anxiety models in rodents. *Psychopharmacology*, 113(3-4), 463-70. [DOI:10.1007/BF02245224] [PMID]
- Guenther, K., Deacon, R. M., Perry, V. H., & Rawlins, J. N. P. (2001). Early behavioural changes in scrapie-affected mice and the influence of dapsone. *European Journal of Neuroscience*, 14(2), 401-9. [DOI:10.1046/j.0953-816x.2001.01645.x] [PMID]
- Hartman, T. K., Rogerson, S. J., & Fischer, P. R. (2010). The impact of maternal malaria on newborns. *Annals of Tropical Paediatrics*, 30(4), 271-82. [DOI:10.1179/146532810X12858955921032] [PMID]

- Iyare, E. E., & Adegoke, O. A. (2008). Mechanism of the delayed puberty onset in offspring of rats that consumed aqueous extract of Hibiscus sabdariffa during pregnancy. *Nigerian Journal of Physiological Sciences*, 23(1-2), 71-7. [PMID]
- Iyare, E. E., & Iyare, F. E. (2006). Effect of inutero exposure to an aqueous extract of hibicus sabdariffa on offspring's postnatal growth in Sprague-Dawley rats. *Nigerian Journal of Health and Biomedical Sciences*, 5(1), 27-30. [DOI:10.4314/njhbs. v5i1.11569]
- Iyare, E. E., & Iyare, F. E. (2007). Maternal consumption of aqueous extract of hibiscus sabdariffa during pregnancy attenuates pregnancy weight gain and postpartum weight loss. *African Journal of Biomedical Research*, 10(3), 257-61. [DOI:10.4314/ ajbr.v10i3.50649]
- McDonald, C. R., Cahill, L. S., Ho, K. T., Yang, J., Kim, H., Silver, K. L., et al. (2015). Experimental malaria in pregnancy induces neurocognitive injury in uninfected offspring via a C5a-C5a receptor dependent pathway. *PLoS Pathogens*, 11(9), e1005140. [DOI:10.1371/journal.ppat.1005140]
- Menendez, C., Romagosa, C., Ismail, M. R., Carrilho, C., Saute, F., Osman, N., et al. (2008). An autopsy study of maternal mortality in Mozambique: The contribution of infectious diseases. *PLoS Medicine*, 5(2), e44. [DOI:10.1371/journal.pmed.0050044]
- Meyer, U., Nyffeler, M., Engler, A., Urwyler, A., Schedlowski, M., Knuesel, I., et al. (2006). The time of prenatal immune challenge determines the specificity of inflammation-mediated brain and behavioral pathology. *Journal of Neuroscience*, 26(18), 4752-62. [DOI:10.1523/JNEUROSCI.0099-06.2006] [PMID]
- Millan, M. J., Girardon, S., Mullot, J., Brocco, M., & Dekeyne, A. (2002). Stereospecific blockade of marble-burying behaviour in mice by selective, non-peptidergic Neurokinin 1 (NK1) receptor antagonists. *Neuropharmacology*, 42(5), 677-84. [DOI:10.1016/S0028-3908(02)00021-7]
- Monif, G. R. G., Baker, D. A. (2004). Infectious disease in obstetrics and gynecology, 6<sup>th</sup> Ed. New York: Parthenon.
- Montegomery, S. A. (1991). Clinical significance of 5-HT reuptake Inhibitors. *Psychopharmacology*, 6(1), S3-7. [DOI: 10.1002/ hup.470060502]
- Neres, R., Marinho, C. R., Gonçalves, L. A., Catarino, M. B., & Penha-Gonçalves, C. (2008). Pregnancy outcome and placenta pathology in Plasmodium berghei ANKA infected mice reproduce the pathogenesis of severe malaria in pregnant women. *PloS One*, 3(2), e1608. [DOI:10.1371/journal. pone.0001608] [PMID]
- Nyffeler, M., Meyer, U., Yee, B. K., Feldon, J., & Knuesel, I. (2006). Maternal immune activation during pregnancy increases limbic GABAA receptor immunoreactivity in the adult offspring: implications for schizophrenia. *Neuroscience*, 143(1), 51-62. [DOI:10.1016/j.neuroscience.2006.07.029] [PMID]
- O'Callaghan, E., Sham, P. C., Takei, N., Murray, G., Glover, G., Hare, E. H., & Murray, R. M. (1994). The relationship of schizophrenic births to 16 infectious diseases. *British Journal of Psychiatry*, 165(3), 353-6. [DOI:10.1192/bjp.165.3.353] [PMID]
- Ofori, M. F., Ansah, E., Agyepong, I., Ofori-Adjei, D., Hviid, L., & Akanmori, B. D. (2009). Pregnancy-associated malaria in a rural community of Ghana. *Ghana Medical Journal*, 43(1), 13-18. [PMID] [PMCID]

- Okojie, A. K., Okobi, O. E., Dar, H., Malik, H., Arif, M., Rauf, K., & Iyare, E. E. (2017). Potential link between Complement 5a Receptor and mood disorders in mouse exposed to experimental malaria In-utero. *Nigerian Journal of Physiological Sciences*, 32(2), 171-7. [PMID]
- Ozawa, K., Hashimoto, K., Kishimoto, T., Shimizu, E., Ishikura, H., & Iyo, M. (2006). Immune activation during pregnancy in mice leads to dopaminergic hyperfunction and cognitive impairment in the offspring: a neurodevelopmental animal model of schizophrenia. *Biological Psychiatry*, 59(6), 546-54. [DOI:10.1016/j.biopsych.2005.07.031] [PMID]
- Pellow, S., Chopin, P., File, S. E., & Briley, M. (1985). Validation of open: Closed arm entries in an elevated plus-maze as a measure of anxiety in the rat. *Journal of Neuroscience Methods*, 14(3), 149-67.[DOI:10.1016/0165-0270(85)90031-7]
- Pinel, J. P., & Treit, D. (1978). Burying as a defensive response in rats. *Journal of Comparative and Physiological Psychology*, 92(4), 708-12. [DOI:10.1037/h0077494]
- Procianoy, R. S., Koch, M. S., & Silveira, R. C. (2009). Neurodevelopmental outcome of appropriate and small for gestational age very low birth weight infants. *Journal of Child Neurology*, 24(7), 788-94. [DOI:10.1177/0883073808331087] [PMID]
- Shi, L., Fatemi, S. H., Sidwell, R. W., & Patterson, P. H. (2003). Maternal influenza infection causes marked behavioral and pharmacological changes in the offspring. *Journal of Neuroscience*, 23(1), 297-302. [DOI:10.1523/JNEUROSCI.23-01-00297.2003]
- Thomas, A., Burant, A., Bui, N., Graham, D., Yuva-Paylor, L. A., & Paylor, R. (2009). Marble burying reflects a repetitive and perseverative behavior more than novelty-induced anxiety. *Psychopharmacology*, 204(2), 361-73. [DOI:10.1007/s00213-009-1466-y] [PMID] [PMCID]
- Umbers, A. J., Aitken, E. H., & Rogerson, S. J. (2011). Malaria in pregnancy: Small babies, big problem. *Trends in Parasitology*, 27(4), 168-75. [DOI:10.1016/j.pt.2011.01.007] [PMID]
- Walsh, R. N., & Cummins, R. A. (1976). The open-field test: A critical review. *Psychological Bulletin*, *83*(3), 482-504. [DOI:10.1037/0033-2909.83.3.482] [PMID]
- Wicke, K. M., & Gross, G. (2005). Marble burying behavior is prevented by anxiolytics as well as by motorstimulants. *Phar*macopsychiatry, 38(5), A253. [DOI:10.1055/s-2005-918875]

This Page Intentionally Left Blank